Author's response to reviews

Title: An adult patient with Nijmegen Breakage Syndrome and Hodgkins Lymphoma

Authors:

Katharina Engel (katharina.engel@lrz.tu-muenchen.de)
Martina Rudelius (rudelius@lrz.tu-muenchen.de)
Felix G Meinel (felix.meinel@med.uni-muenchen.de)
Christian Peschel (christian.peschel@lrz.tu-muenchen.de)
Ulrich Keller (Ulrich.Keller@lrz.tu-muenchen.de)

Version: 3
Date: 26 December 2013

Author's response to reviews: see over
Dear Editor,

please find attached the re-submission of our clinical case report of Nijmegen Breakage Syndrome and Hodgkins Lymphoma in an adult patient.

As requested in your email of November 26, 2013, we have now added all authors email addresses on the first page of the manuscript. Secondly, we have clarified the informed consent for the publication of this case report: on the one hand, it was obtained from the patient himself as he is – despite his mild mental retardation – able to make intelligent decisions of such kind. On the other hand, during his stay in our hospital he was legally represented by his parents and therefore, informed consent was also obtained from them.

I do hope to have answered your concerns and am very much looking forward to hearing from you. Please let me know if you have any further questions.

Sincerely,

Katharina Engel.